Author's response to reviews

Title: Detection of a divergent Parainfluenza 4 in an adult patient with influenza like illness using next-generation sequencing

Authors:

Seweryn Bialasiewicz (seweryn@uq.edu.au)
Jodie McVernon (j.mcvernon@unimelb.edu.au)
Terry Nolan (t.nolan@unimelb.edu.au)
Stephen B Lambert (sblambert@uq.edu.au)
Guoyan Zhao (GZhao@pathology.wustl.edu)
David Wang (davewang@borcim.wustl.edu)
Michael D Nissen (Michael_Nissen@health.qld.gov.au)
Theo P Sloots (t.sloots@uq.edu.au)

Version: 2
Date: 5 February 2014

Author's response to reviews:

As requested by the BioMed Central Journal Editorial Office, the following changes were made the the main body of the manuscript:

1) A conclusion summarising the study's key points
2) email addresses for all authors on the Title page